

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Hoang, Diem

Working Paper Export growth and demographic changes: Evidence from Vietnam

Ruhr Economic Papers, No. 1145

Provided in Cooperation with: RWI – Leibniz-Institut für Wirtschaftsforschung, Essen

Suggested Citation: Hoang, Diem (2025) : Export growth and demographic changes: Evidence from Vietnam, Ruhr Economic Papers, No. 1145, ISBN 978-3-96973-328-8, RWI - Leibniz-Institut für Wirtschaftsforschung, Essen, https://doi.org/10.4419/96973328

This Version is available at: https://hdl.handle.net/10419/314409

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



RUHR ECONOMIC PAPERS

Diem Hoang





Imprint

Ruhr Economic Papers

Published by

RWI – Leibniz-Institut für Wirtschaftsforschung Hohenzollernstr. 1-3, 45128 Essen, Germany Ruhr-Universität Bochum (RUB), Department of Economics Universitätsstr. 150, 44801 Bochum, Germany Technische Universität Dortmund, Department of Economic and Social Sciences Vogelpothsweg 87, 44227 Dortmund, Germany Universität Duisburg-Essen, Department of Economics Universitätsstr. 12, 45117 Essen, Germany

Editors

Prof. Dr. Thomas K. Bauer RUB, Department of Economics, Empirical Economics Phone: +49 (0) 234/3 22 83 41, e-mail: thomas.bauer@rub.de Prof. Dr. Ludger Linnemann Technische Universität Dortmund, Department of Business and Economics Economics – Applied Economics Phone: +49 (0) 231/755-3102, e-mail: Ludger.Linnemann@tu-dortmund.de Prof. Dr. Volker Clausen University of Duisburg-Essen, Department of Economics International Economics Phone: +49 (0) 201/1 83-3655, e-mail: vclausen@vwl.uni-due.de Prof. Dr. Ronald Bachmann, Prof. Dr. Almut Balleer, Prof. Dr. Manuel Frondel, Prof. Dr. Ansgar Wübker RWI, Phone: +49 (0) 201/81 49-213, e-mail: presse@rwi-essen.de

Editorial Office

Sabine Weiler

RWI, Phone: +49 (0) 201/81 49-213, e-mail: sabine.weiler@rwi-essen.de

Ruhr Economic Papers #1145

Responsible Editor: Volker Clausen

All rights reserved. Essen, Germany, 2025

ISSN 1864-4872 (online) - ISBN 978-3-96973-328-8

The working papers published in the series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editors.

Ruhr Economic Papers #1145

Diem Hoang

Export Growth and Demographic Changes: Evidence from Vietnam



Bibliografische Informationen der Deutschen Nationalbibliothek

The Deutsche Nationalbibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data are available on the Internet at http://dnb.dnb.de

RWI is funded by the Federal Government and the federal state of North Rhine-Westphalia.

https://dx.doi.org/10.4419/96973328 ISSN 1864-4872 (online) ISBN 978-3-96973-328-8 Diem Hoang*

Export Growth and Demographic Changes: Evidence from Vietnam

Abstract

This paper evaluates the impacts of trade liberalization on the marital and fertility choices of women in Vietnam. Applying a regional exposure approach, we leverage the U.S.-Vietnam Bilateral Trade Agreement (BTA) as an exogenous and positive shock to the nation's export growth. Our results indicate that young women (aged 18-28) in provinces more exposed to export tariff reductions tend to delay marriage and childbirth. In contrast, we observe no significant impact on the marriage and fertility decisions of older women, nor any changes in sex-selective behavior across the general population. Further analysis reveals that this trade shock does not influence women's participation in the labor force or their employment status, nor does it lead to increased gender segregation in the labor market. The observed delay in marriage and fertility among young women may be attributed to a shift from agriculture to manufacturing and women staying longer at schools.

JEL-Codes: F61, J13, J16

Keywords: Trade liberalization; U.S.-Vietnam BTA; marriage; fertility; sex selection

February 2025

^{*} Diem Hoang, University of Duisburg-Essen. – I would like to thank Martin Karlsson, Brian McCaig, and members of the Research and Training Group 2484 on Regional Disparities and Economic Policies for helpful comments and feedback. – All correspondence to: Diem Hoang, University of Duisburg-Essen, Chair of Health Economics, Weststadttürme, Berliner Platz 6–8, 45127 Essen, Germany, e-mail: hoang.diem@uni-due.de

1 Introduction

Since its establishment in 1947, the World Trade Organization (WTO) and other international organizations have consistently advocated for trade liberalization - the removal of barriers to the flow of goods and services - as a strategy to promote economic prosperity, especially in developing countries (Siddiqui, 2015; Engel et al., 2021). Most research focuses on the economic impacts of integration, such as growth, poverty reduction, employment, skill premiums, or firms' productivity. However, over the past decade, there has been growing concern regarding the role of trade liberalization in driving social and demographic changes in both developed and emerging economies, primarily through the labor market channel (Chakraborty, 2015; Do et al., 2016; Kis-Katos et al., 2018; Anukriti and Kumler, 2019; Giuntella et al., 2022; Keller and Utar, 2022; Mansour et al., 2023; Luo and Zou, 2024; Feng et al., 2024). The findings of these studies are mixed, depending on the nature of liberalization, how it affects men and women in the labor market, and on other factors such as welfare policies and culture. In developing Asia, the persistence of son preference, combined with declining fertility driven by rapid economic growth and family planning policies, is exacerbating the issue of sex imbalance (Das Gupta and Mari Bhat, 1997; Jayachandran, 2017; Babiarz et al., 2018).

This paper provides an insight into the impacts of trade liberalization on social and demographic changes in Vietnam, a small and developing country with a deeply ingrained preference for sons. To establish causal relationships, this paper leverages an exogenous and positive shock to the export performance of Vietnam in the early 2000s: the implementation of the United States and Vietnam Bilateral Trade Agreement (BTA) in December 2001. The U.S.-Vietnam BTA marked a breakthrough in the relationship between the two countries and opened the door for Vietnam to better integrate into the global economy (Riedel et al., 2003). From virtually no trade relation, the U.S. has become Vietnam's largest export market, accounting for nearly 30% of Vietnam's total exports in 2021.¹ This created a significant and exogenous shock to the labor market of Vietnam (McCaig, 2011; Fukase, 2013; McCaig and Pavcnik, 2018).² When the shock occurred, Vietnam's socio-economic conditions had several notable features. First, the country already had a very high level of labor force participation among both men and women.³ In 1999, nearly 80% of women aged 15-64 were active in the labor market.⁴ Second, after a prolonged period implementing family planning policies,

¹Data compiled by WITS (World Integrated Trade Solution).

 $^{^2 \}rm More$ discussion about the U.S.-Vietnam BTA in Section 3.

³Banerji et al. (2018) accredit the high rate of female labour force participation to Vietnam's long period under war (during which time the women had to fill in for the shortage of men in the labor market), and socialism, which stresses equality among all individuals.

⁴Authors' calculation from the Vietnam Population Census in 1999 (the 3% sample).

Vietnam's fertility rate had dropped to a low and stable level of just under two children per woman.⁵ Third, since the early 2000s, the male-to-female sex ratio at birth in Vietnam began to increase abnormally to more than 110 boys over 100 girls (see Figure 1), placing Vietnam second only to China in terms of sex imbalance at birth.⁶ These facts raise the question of to what extent the shock induced by the U.S.-Vietnam BTA has affected Vietnam's demographic structure concerning marriage, fertility, and the sex ratio at birth.

To answer this question, we apply a regional exposure approach following Topalova (2010) and employ data from the Vietnam population census from 1999 to 2019. The indicator of regional exposure to this trade shock is measured by the change in the tariff rate that the U.S. applied on imports from Vietnam in each industry before and after BTA came into effect, interacting with the original share of labor in the corresponding industry in each province. The validity of the approach comes from the exogeneity of the shock (Borusyak et al., 2022), as tariff reduction was implemented immediately and unrelated to the performance of the U.S. or Vietnam economies (McCaig, 2011; McCaig and Pavcnik, 2018). Our results show that provinces more exposed to tariff cuts experienced a reduction in marriage and fertility among young women (aged 18-28) but not among older women (aged 29-49), implying a delay or postponement of marriage and fertility. When distinguishing the trade shock by gender intensity in the initial industry structure of each province, we find that most of the effects are driven by female-intensive shocks. Mediation analysis suggests that these effects are driven by structural change, where women in provinces more exposed to tariff reductions are more likely to move from agriculture to manufacturing, which could raise the opportunity cost of marriage and having children (Erten et al., 2023; Doepke, 2004), and by women staying longer at school. However, we do not find evidence of the BTA's impact on gender allocation in the labor market (i.e., no impacts on the gender intensity of the industry structure at the provincial level or on the indicator of gender segregation in the labor market). This might explain why, unlike the case of India (Chakraborty, 2015) or China (Luo and Zou, 2024), we do not find significant impacts of trade liberalization on the child sex ratio in Vietnam.

Our paper contributes to the growing body of literature on the impacts of economic integration on social and demographic transformations. Most studies in this area have focused on the effects of increasing import competition, either due to China's rising prominence in the global market or domestic trade reforms aimed at reducing protectionist barriers. In the former case, Autor et al. (2019) find that increased import

⁵Starting in the 1960s in the North and since 1975 in the South, the Vietnamese Government promoted a model of small families with two to three children. In 1988, they officially issued a policy limiting the number of children to two per family. As a result, Vietnam's fertility rate declined significantly from 6 to just 2 children per woman from the 1960s to the early 2000s (Ngo, 2020).

⁶Source: https://liveprod.worldbank.org/en/indicator/sp-pop-brth-mf.

competition has adverse effects on both the *absolute* and *relative* employment and earnings of young men in the U.S., which subsequently leads to lower marriage and fertility rates. Keller and Utar (2022) identify parallel negative short-term impacts on men and women in Denmark, but women are more likely to marry or have children. In developing countries, Braga (2018) and Mansour et al. (2023) report that labor market outcomes are more negatively affected for women than for men in Brazil and Chile, respectively. While Braga (2018) observes adverse effects on fertility rates but no significant impact on marriage or cohabitation behaviors, Mansour et al. (2023) finds a decline in marriage rates without a corresponding impact on fertility. Reducing input tariffs has improved labor force participation and employment prospects of low-caste or low-skilled women (relative to men) in India and Indonesia, respectively (Anukriti and Kumler, 2019; Kis-Katos et al., 2018). While Kis-Katos et al. (2018) find lower marriage and fertility rates among young women, Anukriti and Kumler (2019) report a higher fertility rate among low-caste women in rural areas. Studies on the impacts of export growth mainly focus on China, where lower tariffs on Chinese exports are found to improve the economic condition of women relative to men, which contributes to the postponement of marriage and fertility among young women (Luo and Zou, 2024). In the context of low fertility, a worse employment prospect induced by higher import competition with Eastern Europe is associated with lower fertility in Germany (Giuntella et al., 2022). At the same time, structural transformation driven by foreign direct investment in China is found to lower marriage and fertility (on the extensive and intensive margin) of women (Erten et al., 2023). Overall, findings from the literature are mixed, and many of the results are explained by the gender differential impacts of trade liberalization on labor market outcomes. Our paper offers insights into the impacts of export growth in a developing country with an exceptionally high level of female labor force participation *priori* (which was higher than those in the countries mentioned above) and a relatively low fertility rate. Additionally, the effects of the positive trade shock in Vietnam appear to be relatively gender-neutral.

Our paper also contributes to the limited literature on the impacts of trade liberalization on the pre- or post-natal sex selection behavior (Chakraborty, 2015; Anukriti and Kumler, 2019; Luo and Zou, 2024). The phenomenon of sex selection is prevalent in countries with a persistent preference for sons, such as China, India, and Vietnam. Anukriti and Kumler (2019) finds that the positive impacts of trade liberalization in India on employment and fertility of low-caste women are associated with a higher fraction of newborn females and a lower mortality rate for girls within the first year of life. Similarly, the narrower gender gap in the labor market induced by trade shock in China is associated with a higher female-to-male ratio among newborn children, though it only delays the timing of births for women (Luo and Zou, 2024). In Vietnam, an absence of changes in the fertility rate or relative employment of women might help explain the null effect of trade liberalization on the child sex ratio. Another driving factor is the fact that the fertility rate in Vietnam is higher than in China, and most parents prefer to have mixed-sex children (Vu, 2014; Vu et al., 2021).

Our paper extends the existing empirical research on the impact of the U.S.-Vietnam BTA, which has predominantly focused on economic outcomes. Short-term analyses show the positive impacts of BTA on poverty reduction (McCaig, 2011), skill premiums (Fukase, 2013), labor reallocation from the informal to the formal sector, and firm productivity (McCaig and Pavenik, 2018). Long-term analyses highlight the role of BTA in facilitating structural transformation (Asghar and McCaig, 2024) and attracting foreign direct investment in sectors experiencing higher tariff cuts (McCaig et al., 2022). Our analysis complements two recent studies, which use different data sets to evaluate the impact of BTA on marriage and fertility (Ngoc et al., 2024) and on the sex ratio (Huynh and Nguyen, 2023). Ngoc et al. (2024) find that BTA reduces marriage rates and fertility among women under 40 but increases marriage rates and the number of children among older women without proper explanation. Huynh and Nguyen (2023) find a lower incidence of childbirth and a higher fraction of boys born to married couples. Our study utilizes more comprehensive data than these studies, specifically the population census, which includes information on the total number of births a woman has had at the time of the survey.⁷ In our analysis, we can investigate the impacts of BTA on women across different age groups, providing clearer insights into whether BTA reduces or delays women's marriage and fertility. We also reveal that most of these effects are driven by shocks to female-intensive industries.

The paper is organized as follows: Section 2 synthesizes the channels through which trade liberalization affects marriage, fertility, and sex selection. Section 3 briefly introduces the context and commitments within the U.S-Vietnam Bilateral Trade Agreement (BTA). Section 4 describes the data used in the analysis and presents some general descriptive statistics. Section 5 outlines the empirical approach. Section 6 discusses the main results. Section 7 explores potential mechanisms. Finally, Section 8 concludes the paper.

⁷Ngoc et al. (2024) exploit the Vietnam Multiple Indicator Cluster Surveys (MICS) for their analysis. The MICS sampling was designed to represent national, regional, and rural/urban areas. Despite the authors applying the provincial exposure approach, it does not necessarily assure representativeness at the provincial level. When analyzing the impact of BTA on marriage and fertility among women under 40, the authors use cross-sectional data from the 2013/2014 survey round and create retrospective panel data to analyze the dynamic impacts of the shock. This approach further questions the representativeness of the data at the provincial level and over time while implicitly assuming no migration during the study period. Huynh and Nguyen (2023) use data from the Vietnam Household Living Standard Surveys (VHLSS) to assess the impacts of BTA on fertility and sex selection. While VHLSS data are representative at the provincial level, the authors only consider households where both parents are working. Additionally, VHLSS only provides information about living and co-residing children, which does not always reflect the actual number of births by the mothers.

2 Conceptual framework

Classical trade theory emphasizes the benefit of lowering barriers to the flow of goods and services across countries through improved productivity and reduced consumer prices. David Ricardo's trade theory postulates that trade encourages countries to specialize in the industries in which they have a comparative advantage. Therefore, these industries will expand, while the sectors with comparative disadvantage will contract (Ricardo, 2005). While the Ricardian model explains the comparative advantage by the difference in technologies, the Heckscher-Ohlin model derives such advantage from the difference in factor endowment (capital, labor, land) and factorintensity of goods (Morrow, 2010).⁸ As a consequence, trade liberalization plays an important role in driving the process of structural transformation in both developed and developing countries (i.e., China experiences massive structural change from agriculture to manufacturing and services sectors following its accession to the World Trade Organization (Erten and Leight, 2021)). In a scenario of lowering tariffs between an advanced economy and a developing economy, these models predict that the latter would benefit from the expansion of low-skilled, labor-intensive manufacturing sectors, which would increase income and employment for the general population (as this country is relatively labor-abundant).

The effects of trade liberalization depend on whether it is associated with lowering domestic protection (i.e., reduced import tariffs on final goods) or opening international markets (i.e., reduced export tariffs by trading partners). In the former case, domestic producers would face higher competition from abroad, resulting in job loss or income decline in sectors with weak competitiveness.⁹ In the latter case, the country might experience export growth with the overall net impacts on growth and income. In addition, the impacts of trade liberalization depend a lot on the initial structure of the economy. If there are also gender differences in the allocation of labour across sectors, it also affects the gender disparities with respect to income and employment (Brussevich, 2018; Besedeš et al., 2021; Gupta, 2021; Juhn et al., 2013).¹⁰ This, in turn, affects other non-economic outcomes, such as marriage or fertility (Do et al., 2016; Autor et al., 2019).¹¹ The explanation underlying these non-economic impacts lies in the division of labor within a household: men specialize in market work, and women bear most of

⁸However, these theories are based on a static setting and does not consider dynamic adjustments. It also assumes perfect competition, full employment, and free mobility of labors across sectors (Morrow, 2010; Siddiqui, 2018)

⁹For examples, trade liberalization in Brazil in the late 1980s and early 1990s is found to be associated with lower labour force participation and employment of both men and women, especially in traded sector (Gaddis and Pieters, 2017).

¹⁰Note that these studies also emphasize other factors associated with trade liberalization that affect gender inequality (such as technological adjustment).

¹¹Do et al. (2016) theoretically and empirically prove that countries with comparative advantage in female-intensive goods exhibit lower levels of fertility.

the burden of domestic work and child-rearing (Becker, 1981, 1985). An increase in the earnings of men relative to women would increase marriage and fertility, contrary to the relative increase in the earnings of women (Becker and Lewis, 1973; Stevenson and Wolfers, 2007; Do et al., 2016; Autor et al., 2019).

Even if there are no changes in the relative earnings and employment levels between men and women, the overall improvement in incomes and economic opportunities for the entire population would still influence marriage and fertility patterns. The economic analysis of fertility over the last half of a century, pioneered by the work of Becker (1960), has been centered around the framework of quality-quantity trade-off (Doepke, 2015). According to Becker (1960), the level of fertility is determined by factors such as income, cost of raising children, and tastes. Holding other things constant, an increase in income would be associated with an increase in the desire for both the number and quality of children. However, the income elasticity with respect to the quality of children would be larger than that with respect to the quantity of children (Becker and Lewis, 1973). In another aspect, the rising income and employment conditions of women push up the opportunity cost of raising children, again reducing fertility. Analyzing data on developing countries, Chatterjee and Vogl (2018) find that long-run economic growth is often associated with delay and decline in fertility, with faster fertility decline among the younger cohort (between 20-34 years old).

In poor and developing countries, where a well-functioned welfare system is absent, having children is also considered as an investment in old age security (where the children often take care of their parents in their old ages) (Nugent, 1985; Boldrin and Jones, 2002). Empirical evidence shows that the provision or extension of social pension drives down fertility (Nugent, 1985; Ebenstein and Leung, 2010; Rossi and Godard, 2022). This suggests that the impact of economic shocks on fertility would be systematically different between developed and developing countries, and their effects on fertility in developing countries where fertility is already at a low level would be more limited.

Trade liberalization would have an impact on the sex imbalance, particularly in countries with son preference. First, if trade liberalization improves women's economic empowerment (i.e., through the expansion of female-intensive sectors), there would be relatively more demand for girls as the expected returns to girls increase. Hence, girls would have more chances to be born or survive in their early ages (Qian, 2008). However, if son preference is persistent, fertility decline (due to trade liberalization) would be associated with more sex imbalance as the desired number of boys does not change while the desired number of children goes down (Jayachandran, 2017). In addition, higher income enables parents to afford costly sex-selective technologies (such as the In-Vitro Fertilization (IVF) technology that allows parents to select the child's gender

at conception; or the technologies used for very early detection of the child's gender in utero).

3 The Vietnam - U.S. Bilateral Trade Agreement

After the end of the Vietnam War in 1975, the U.S. imposed a trade embargo on Vietnam, effectively prohibiting any economic relationship with Vietnam. Since the mid-1990s, efforts to normalize the two countries' relationship have proved successful, with the lifting of the embargo in 1994. This went further with the approval of the U.S.-Vietnam Bilateral Trade Agreement in 2001, in which the U.S. granted Vietnam a Normal Trade Relations (NTR) or the Most Favored Nation (MFN) status to enter the U.S. market. This marks a milestone in the relationship between the two countries and paved the way for Vietnam to negotiate with other countries to approve its accession to the World Trade Organization (WTO) in 2007 (Riedel et al., 2003).

The U.S.-Vietnam BTA opened up significant opportunities for Vietnam to expand its exports to the U.S. market. Before the agreement, products originating from Vietnam faced extremely high tariff rates, as outlined in Column 2 in the U.S. Harmonized Tariff Schedule (HTSUS), similar to other countries without a normal trade relationship with the U.S, such as North Korea and Cuba. Following the implementation of the BTA in December 2001, Vietnamese products became subject to considerably lower MFN tariffs. The average tariff rates for Vietnamese products fell dramatically from about 40% to around 3-4%. From the Vietnamese side, even before BTA came into effect, Vietnam had already applied the MFN tariff rates on goods and services from the U.S. Most of the commitments of Vietnam, however, involved institutional and policy reforms necessary to classify Vietnam as a market economy (i.e. market access for trade in services, regulations relating foreign direct investment, intellectual property rights, transparency of laws and regulations, etc.) (Riedel et al., 2003).

Figure A.1 in the Appendix shows that the share of the U.S. exports in the total exports of Vietnam to the world market increased substantially, rising from around 7% to nearly 20% in just five years, from 2001 to 2006. The share of manufacturing exports to the U.S. market saw an even greater increase, jumping from around 4% in 2011 to nearly 30% in 2006. It is important to note that the implementation of the U.S.-Vietnam BTA does not lead to the trade diversion of Vietnamese exports from the other markets to the U.S. market. As illustrated in Figure A.2, after 2001, exports from Vietnam to both the U.S. and other markets increased significantly.

4 Data and Descriptives

4.1 Data

The Population Census: This paper utilizes data from three population censuses of Vietnam, conducted in 1999 (3% of the full population), 2009 (14% sample), and 2019 (8% sample). The censuses provide detailed information regarding the household composition (relationship to the household head, ethnicity, rural/urban area) and individual characteristics (age, gender, education, employment, marital status). Regarding economic activity, respondents were asked whether they were employed, unemployed or inactive in the labor market. For those employed, detailed information is available at the 3-digit level of the industrial sector.

A special module was administered to women aged 15 to 49 years, asking about their reproductive lives, including the total number of children they have ever given birth to, the number of children still alive, and details about their most recent birth (year and month of birth, and the number of boys and girls in the last birth). Using this module, we construct the outcome variables for women, focusing on fertility and the gender of the latest birth. For simplicity, we exclude cases of non-singleton last births, which account for less than 2% of all reported last births. This allows us to assess the impact of trade liberalization on sex selection, which often happens at the last birth. However, not all the latest births in our sample are truly the last births, as many women were still very young at the time of the survey. To complement our analysis of the impact of trade liberalization on sex selection, we also performed an analysis at the provincial level, examining changes in the sex ratio among children under five years old in each province. Throughout the paper, we primarily focus on the marriage and fertility behaviors of women aged 18 to 49, as Vietnamese Family Law sets the minimum legal marriage age for women at 18.¹²

Tariff data: We utilize the tariff cuts matched with the 2-digit industrial classification of Vietnam in 1993, as provided by McCaig (2011).¹³ The author calculates the tariff rates applied to Vietnamese products using ad-valorem tariffs weighted by the share of the U.S.'s imports of each industry over its total imports from the world market. A total of 76 industries experienced tariff reductions, with the highest drops in manufacturing sectors, where tariffs declined from an average of 33% to 3.4%.

Table 1 lists the industries with the level of tariff cuts of more than 30% following BTA. These are all manufacturing industries, including garment and textiles,

 $^{^{12}\}mathrm{According}$ to data from the 1999 census, the share of ever-married women among all women aged 10-17 was 0.6%.

¹³The industry codes used the Population Census in 1999 were based on the Vietnam Standard Industrial Classification (VSIC) 1993; The Population Census in 2009 followed the VSIC in 2007; and the Population Census in 2019 followed the VSIC in 2018.

manufacture of tobacco products, furniture, plastics and rubbers, and electric appliances.

Industry	Column 2 tariff	MFN tariff	Change	Industry Classification
code	rate $(\%)$	rate $(\%)$	(%)	
24	35.18	3.19	32.00	Manuf. of chemicals and chemical products
31	35.32	2.81	32.52	Manuf. of electrical machinery and apparatus n.e.c.
29	34.30	1.46	32.84	Manuf. of not-yet-classified machinery and equipment
32	34.88	0.70	34.17	Manuf. of radio, television and communication equipment and apparatus
30	34.99	0.03	34.96	Manuf. of office, accounting and computing machinery
28	39.09	2.93	36.17	Manuf. of fabricated metal products, except machinery and equipment
26	42.25	4.46	37.79	Manuf. of other non-metallic mineral products
33	40.95	1.09	39.86	Manuf. of medical, precision and optical instruments, watches and clocks
25	43.61	3.68	39.93	Manuf. of rubber and plastics products
36	43.16	1.24	41.91	Manuf. of furniture; manufacturing n.e.c.
17	62.67	14.19	48.48	Manuf. of textiles
16	63.83	5.67	58.15	Manuf. of tobacco products
18	74.05	15.53	58.53	Manuf. of wearing apparel; dressing and dyeing of fur

Table 1: Industries with highest level of tariff reductions

Notes: Data are sorted from McCaig (2011)

4.2 Descriptives

Vietnam stands out from many other countries in the region with a very high level of women's labor force participation (Banerji et al., 2018). Table 2 shows that the share of women who engaged in some economic activities was high at approximately 80% with minor changes from 1999 to 2019. Disaggregating by employment sector, we observe a remarkable fall in the share of women working in agriculture, from 68.3% in 1999 to 51.9% in 2009 and 28.9% in 2019. During the same period, the share of women working in the manufacturing sector increased substantially from 10% in 1999 to 21.8% in 2019. The shares of women aged 18-49 who have never married declined from 25.4% in 1999 to 22.4% and 21.8% in 2009 and 2019, respectively. There was also a slight decrease in the average number of children per woman, from 1.98 to 1.44 children per woman in 1999 and 2019, respectively. The probabilities of having a female in the latest birth were lower than the natural level and declined somewhat from 45% in 1999 to 43.8% in 2019.¹⁴ This partly reflects the son-stopping behavior of parents in Vietnam (indicating a higher likelihood of ceasing childbearing after the birth of a son compared to a daughter).

At the provincial level, the shares of girls among children under 5 years old in 2009 and 2019 are all below the level in 1999, suggesting evidence of sex selection.¹⁵ Compared to the level in 1999, the number of girls per 1,000 children aged 0-5 decreased by an average of 8.3 in 2009 and 5.5 in 2019.. Other figures at the aggregate level also imply the transition out of agriculture in Vietnam (the share of working people (aged 15-64 years old) in agriculture decreased from 72.45% in 1999 to 58.90% in 2009

¹⁴The normal sex ratios are between 103 to 106 boys over 100 girls, equivalent to the share of female children between 48.3-49.3% (Tafuro and Guilmoto, 2020).

¹⁵Note that in Vietnam, the male-to-female ratio among deceased children is higher than the world's average (Iqbal et al., 2018).

and 40.5% in 2019), and the urbanization process underway (the share of the total population living in the rural areas fell from 79.5% in 1999 to 71.45% in 2019).

	Census 1999	Census 2009	Census 2019
Sample of women aged 18-49			
% single	25.4	22.4	21.8
no. of children	1.98	1.56	1.44
latest birth is female $(\%)$	45	44.3	43.8
% working	79.1	81	78.4
agriculture	68.3	51.9	28.9
manufacturing	9.9	16.9	28.7
Aggregate data at provincial's le	evel		
% girl among children under-5	48.12	47.47	47.75
% rural population	79.50	75.21	71.45
% labor in agriculture	72.45	58.90	40.50
% labor in manufacturing	7.25	11.11	17.66

Table 2: Descriptive statistics, Census 1999-2019

Notes: Authors' calculation from the Population Census in 1999, 2009, and 2019.

Figure 1 shows the evolution of the total fertility rate¹⁶ and the sex ratio at birth (SRB) of Vietnam, retrieved from World Bank data. The sharp decline in the fertility rate of Vietnam before the 21^{st} century was the result of the relatively stringent implementation of the family planning policy of Vietnam. After that, the fertility rate remains quite stable at just under two children per woman. On the contrary, the SRB of Vietnam was quite stable at around 106 boys over 100 girls, then increased in the early 2000s to a peak of around 112 boys over 100 girls.¹⁷

5 Approach

We follow the typical regional approach to assess the impact of trade liberalization, pioneered by Topalova (2007, 2010), and later widely used in the literature of trade liberalization (Kovak, 2013; Dix-Carneiro and Kovak, 2015, 2017; Autor et al., 2013, 2014, 2019; McCaig, 2011). This approach asserts that each location experiences varying levels of exposure to tariff reduction based on its initial differences in industrial composition. We first compute a measure indicating the extent of location exposure to tariff cuts. In our paper, we perform the analysis at the provincial level, the second level of the administrative division of Vietnam. The indicator of provincial exposure to tariff reduction, $\Delta \tau_p$, is calculated as follows:

¹⁶The fertility rate is calculated as the average number of children per woman aged 15-49.

¹⁷This increase was mainly driven by sex-selective behaviour with the dissemination of ultrasound machines and the approval of safe abortion practices in Vietnam in the early 2000s. Post-natal sex selection in Vietnam was rare because, during the period when Vietnam strictly implemented the two-child policy (i.e., from 1988 until the late 1990s), the child sex ratios (from ages 0 to 10) still lay in the normal range.



Figure 1: VN's fertility and sex ratio at birth, 1980-2022 Source: World Bank data.

$$\Delta \tau_p = \sum \frac{L_{ip_0}}{L_{p_0}} \Delta \tau_i \tag{1}$$

In equation 1, p denotes province, i denotes industry. $\frac{L_{ip}}{L_p}$ is the share of workers in industry i in the total employment of province p at based-line (year 1999). $\Delta \tau_i$ is the change in the tariff rate imposed on industry i, which is calculated by taking the tariff rate before 2001 (Column 2) minus the tariff rate after 2001 (MFN tariff).¹⁸ This approach assigns null tariff change to the non-traded sector. Kovak (2013) provides a detailed discussion on constructing the measure of regional exposure to trade liberalization. He suggests removing the non-traded sectors and weighing the level of tariff change by the share of employment in the traded sectors only. However, as he points out, this approach is mostly applied to studies on wage-related outcomes, while our focus is more on non-wage outcomes (marriage, fertility, sex selection). Therefore, we also consider the traded sector's size relative to the total employment of the province at the baseline, as the approach in Autor et al. (2019). Figure A.3 illustrates the variation in the level of exposure to tariff reductions across provinces in Vietnam.

In the main analysis, we investigate the changes in the outcomes at the individual level. Our model is as follows:

$$Y_{ipt} = \sum_{s=2009,2019} \beta_t \Delta \tau_p \mathbf{1}(t=s) + \delta X_{ipt} + \eta X_{p_0t} + \theta_p + \gamma_t + \epsilon_{ipt}$$
(2)

In which, Y_{ipt} is the outcome of woman *i* living in province *p* at the time of survey *t*. We interact the level of provincial exposure to tariff cut with dummies indicating whether *t* equals 2009 or 2019, with 1999 serving as the reference year. We include in

¹⁸in percentage points, i.e. $\Delta \tau_i = 20$ percentage points if tariff in industry *i* dropped from 30% to 10%.

the estimation a set of person *i*'s characteristics (X_{ipt}) , including age and age-squared. X_{p_0t} is a set of baseline characteristics of the province (in 1999) and interacted with a linear time trend. These include the share of women working in agriculture and manufacturing, the share of women (aged 18-49) who have never married, and the share of the population living in rural areas. We also include province (θ_p) and survey time (γ_t) fixed effects.

Following Autor et al. (2019), we also calculate the measures of provincial exposure to tariff cuts differentiated by gender. As discussed in Section 2, the impacts of trade liberalization also depend on the initial gender allocation across sectors. The locations with more penetration of female workers in industries that experience large tariff cuts would expect larger impacts on women. The composite measure of provincial exposure to tariff cut is apportioned into two additive components: locational female industry shock $(\Delta \tau_p^f)$ and male industry shock $(\Delta \tau_p^m)$. The formulas are as follows:

$$\Delta \tau_p^f = \sum \frac{f_{ip_0} \times L_{ip_0}}{L_{p_0}} \Delta \tau_i \tag{3}$$

$$\Delta \tau_p^m = \sum \frac{(1 - f_{ip_0}) \times L_{ip_0}}{L_{p_0}} \Delta \tau_i \tag{4}$$

In which f_{ip_0} is the share of female workers in sector *i* in province *p* in the baseline year (1999), and $1 - f_{ip_0}$ is the corresponding share of male workers. We then estimate a similar model like equation 2, replacing $\Delta \tau_p$ by $\Delta \tau_p^f$ and $\Delta \tau_p^m$ as follows:

$$Y_{ipt} = \sum_{s=2009,2019} \beta_t^f \Delta \tau_p^f \mathbf{1}(t=s) + \sum_{s=2009,2019} \beta_t^m \Delta \tau_p^m \mathbf{1}(t=s) + \delta X_{ipt} + \eta X_{p_0t} + \theta_p + \gamma_t + \epsilon_{ipt}$$
(5)

Table 3 presents the summary statistics of $\Delta \tau_p$, $\Delta \tau_p^f$, and $\Delta \tau_p^m$. The average provincial exposure to export tariff reduction is 7.93 percentage points, with a standard deviation of 1.37 percentage points. The means of provincial male and female trade shock are quite similar and higher for females (4.06 percentage points) than males (3.85 percentage points).

Table 3: Provincial's exposure to tariff reduction

	mean	(std.)	min	\max	Ν
$\Delta \tau_p$	7.93	(1.37)	5.95	14.75	60
$\Delta \tau_p^f$	4.06	(0.85)	2.83	8.37	60
$\Delta \tau_p^m$	3.85	(0.57)	3.00	6.37	60

Notes: Summary statistics of the measures of provincial's exposure to tariff reduction following the formulas in 2, 3, and 4.

Our outcomes of interest include a dummy indicating whether the woman is single or ever-married, the number of children she has ever given birth to, and a dummy equal to one if the gender of the latest birth of the woman is a boy. As the latter outcome is not so informative about sex selection behavior, we complement the analysis at the provincial level. We construct an outcome equal to the share of girls among under-5-year-old children.¹⁹ The estimation at the provincial level is as follows:

$$Y_{pt} = \sum \beta_t \Delta \tau_p \mathbf{1}(t = 2009 | t = 2019) + \eta X_{p_0 t} + \theta_p + \gamma_t + \epsilon_{ipt}$$
(6)

Where Y_{pt} is the outcome of province p at time t. We only include in the estimation the baseline characteristics of the province as in model 2, and add the share of girls among children under 5 in each province in 1999 (interacted with a linear time trend).

This approach can only yield a consistent estimate of β_t in equation 2, 5, and 6 if the measure of regional exposure to tariff-cut does not correlate with any unobserved factors that also affect outcomes (such as technological changes or any other industrial shocks). Goldsmith-Pinkham et al. (2020) argue that this approach can still yield consistent estimates if the "shares" (the initial structure of labor allocation) are exogenous and examining how different exposure to common shock affects the *changes* in the outcomes (the "share" can still be correlated with the *levels* of the outcomes). Borusyak et al. (2022) instead exploit the exogeneity of the shocks themselves while allowing endogenous exposure to shocks. In our setting, the "shocks" exogeneity condition is more likely to be satisfied than the exogenous "share" condition. McCaig (2011) and McCaig and Pavcnik (2018) have performed exhaustive exercises to prove the exogeneity of the tariff cuts following the implementation of the U.S.-Vietnam BTA (i.e., (a) both the tariff rates in Column (2) or the MFN rates are exogenous to the economic condition of Vietnam; and the tariffs in Column (2) are more politically than economically determined; (b) there are no correlations between tariff cuts and the previous trends in the exports from Vietnam to the United States or from Vietnam to other destinations).

6 Results

6.1 Main results

6.1.1 Marriage and fertility

Table 4, Panel A, presents the results of the main estimations regarding the impact of trade liberalization on marriage and fertility of women aged 18-49 years old. The first four columns examine the probability of being single (never married) across different age groups. For the sample of all women (column (1)), the estimated coefficients are

 $^{^{19}\}mathrm{We}$ will discuss more about this in Section 6.1.2.

positive and statistically significant at the 1% level. A one percentage point increase in the provincial exposure to tariff cuts is associated with an increase of 1.1 and 1.6 percentage points in the probability of being single in 2009 and 2019, respectively. When disaggregating by age groups, the impacts of the U.S-Vietnam BTA are primarily observed among young women aged 18-28 years. An increase of one percentage point in the exposure to tariff cuts is associated with 2.0 and 2.8 percentage points increases in the probability of being single in 2009 and 2019 among these youngest women, respectively. There is only a statistically significant impact of BTA on the likelihood of remaining single among women aged 29-39 in 2019, but not in 2009. The estimates for older women (aged 40-49) are positive but insignificant. These results show evidence of delayed marriage among more exposed provinces over time.

The impacts of BTA on fertility rate are presented in columns (5) to (8). All estimated coefficients are negative but not statistically significant except for the estimates on the sample of women aged 18-28. An increase of 1% in the exposure to tariff cuts reduced the average number of children among the youngest women by 0.032 and 0.044. These results again suggest young women delay having children, which is consistent with the pattern of delaying marriage. However, BTA has no impact on the fertility rate of older women.

	being single				no. of children			
	18-49	18-28	29-39	40-49	18-49	18-28	29-39	40-49
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Panel A								
$\Delta \tau_p \times 2009$	0.011^{***}	0.020^{***}	0.006	0.005	-0.018	-0.032***	-0.010	-0.010
	(0.004)	(0.003)	(0.005)	(0.004)	(0.033)	(0.011)	(0.047)	(0.065)
$\Delta \tau_p \times 2019$	0.016^{***}	0.028^{***}	0.013^{**}	0.006	-0.015	-0.044***	-0.010	0.024
	(0.005)	(0.004)	(0.006)	(0.004)	(0.037)	(0.011)	(0.050)	(0.074)
Panel B								
$\Delta \tau_p^f \times 2009$	0.030^{***}	0.055^{***}	0.013	0.016^{**}	-0.080	-0.126^{***}	-0.086	0.041
	(0.009)	(0.014)	(0.010)	(0.007)	(0.063)	(0.036)	(0.089)	(0.132)
$\Delta \tau_p^f \times 2019$	0.022^{**}	0.047^{***}	0.016	0.010	-0.034	-0.103***	-0.040	0.119
	(0.009)	(0.013)	(0.011)	(0.008)	(0.081)	(0.034)	(0.116)	(0.160)
$\Delta \tau_p^m \times 2009$	-0.009	-0.017	0.001	-0.009	0.022	0.081	0.042	-0.174
	(0.011)	(0.020)	(0.010)	(0.007)	(0.066)	(0.049)	(0.102)	(0.163)
$\Delta \tau_p^m \times 2019$	0.019	0.021	0.014	0.003	-0.046	0.009	-0.029	-0.206
1	(0.013)	(0.022)	(0.013)	(0.010)	(0.109)	(0.048)	(0.160)	(0.231)
			. ,					
Mean Y	0.223	0.469	0.086	0.054	1.636	0.651	1.973	2.601
Ν	6,070,197	$2,\!261,\!477$	$2,\!157,\!007$	$1,\!651,\!713$	$6,\!070,\!197$	$2,\!261,\!477$	$2,\!157,\!007$	$1,\!651,\!713$

Table 4: Impacts of BTA on marriage and fertility

Notes: the estimations also include the women's age, age-squared, ethnicity, provinces' based line characteristics in 1999 interacted with a linear time trend, province and census year fixed effects. Standard errors are clustered at the provincial level and in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01.

Panel B of Table 4 presents the results with gender-specific trade shocks. It appears that most of the effects on women's marriage and fertility are driven by female-industry shock. A one standard deviation (0.85 percentage points) increase in the province's exposure to female trade shock is associated with 2.6 (0.03×0.85) and 1.9 (0.022×0.85) percentage points increase in the probability of being single for women in 2009 and 2019, respectively. The effects are mainly concentrated among the youngest group. A one standard deviation increase in the province's exposure to female trade shock is associated with a 4.7 and 4.0 percentage points increase in the likelihood of never marrying among women aged 18-28 in 2009 and 2019, respectively. The same pattern was also observed for fertility outcomes. Female-intensive trade shock does not affect the overall level of fertility of all women but reduces the number of children among women aged 18-28. A one standard deviation increase in the province's exposure to female-intensive shock is associated with a reduction in the average number of children among the youngest women by 0.107 in 2009 and by 0.089 in 2019.

6.1.2 Sex selection

Table 5 presents the estimates of the impacts of BTA on sex selection. The analysis at an individual level (Columns (1) to (4)) shows no statistically significant impacts of the trade shocks (and gender-specific shocks) on the probability of having a girl at the latest birth. However, the analysis at the individual level might be only informative for the youngest group of women. The latest birth among this group is more likely to be the first or second birth, while it is more likely to be the last birth among the oldest women. In all periods with and without sex-selective technologies, a large majority of parents still follow son-stopping behaviour: they continue childbearing until they have at least one male child. Therefore, the old mothers (aged 40-49) in 1999 (the period when sex-selective technologies were inaccessible) could still have an equal or higher probability of having a boy in the last birth compared to the old mothers in 2009 or 2019 (the periods in which sex-selective technologies were widely available in Vietnam). In Vietnam, there is evidence of sex selection even from the first birth, as the sex ratio of first-order births is higher than that of the second-order births (UNFPA, 2014, 2019). Therefore, the younger mothers (aged 18-28 and 29-39) in the period with sex-selective technologies (in 2009 and 2019) would have a higher probability of having a boy in the latest birth (due to sex-selection) compared to the mothers of the same ages in 1999. Even so, we do not observe statistically significant impacts of BTA on the sex ratio among the latest births of these younger mothers (though the coefficients are negative in Column (2) Panel (A)).

The results of the estimations at the provincial level (column (5)) corroborate our conclusion from the estimates on individual women. In any specifications, we do not observe statistically significant impacts of the trade shock on the sex ratio among children under 5 years old in each province. However, it should be noted that the coefficients in Panel (A) and the coefficients of the female-intensive shocks in Panel

		provincial level			
	18-49	18-28	29-39	40-49	
	(1)	(2)	(3)	(4)	(5)
Panel A					
$\Delta \tau_p \times 2009$	-0.0003	-0.0019	0.0014	-0.0015	-0.0008
-	(0.0017)	(0.0020)	(0.0012)	(0.0031)	(0.0010)
$\Delta \tau_p \times 2019$	0.0007	-0.0007	0.0014	0.0001	-0.0013
	(0.0020)	(0.0025)	(0.0013)	(0.0033)	(0.0012)
Panel B					
$\Delta \tau_p^f \times 2009$	0.0023	-0.0006	-0.0003	0.0087	-0.0023
r	(0.0040)	(0.0055)	(0.0033)	(0.0062)	(0.0023)
$\Delta \tau_p^f \times 2019$	0.0036	0.0056	-0.0020	0.0107	-0.0010
1	(0.0050)	(0.0066)	(0.0035)	(0.0079)	(0.0028)
$\Delta \tau_p^m \times 2009$	-0.0045	-0.0041	0.0039	-0.0171^{*}	0.0016
	(0.0058)	(0.0079)	(0.0060)	(0.0087)	(0.0034)
$\Delta \tau_p^m \times 2019$	-0.0039	-0.0109	0.0069	-0.0163	-0.0019
	(0.0076)	(0.0094)	(0.0067)	(0.0120)	(0.0037)
$\operatorname{Mean}_{N} \mathbf{Y}$	0.444	0.464	0.444	0.429 1 514 338	0.478
± •	1,100,000	$\pm,000,022$	1,010,200	1,011,000	100

Table 5: Impacts of BTA on probability of having a female child in the latest birth

Notes: the estimations also include the women's age, age-squared, ethnicity, province's based line characteristics in 1999 interacted with a linear time trend, province and census year fixed effects. Standard errors are clustered at province level and in parentheses, and with bootstrap procedure in Column (5). *p < 0.10, **p < 0.05, ***p < 0.01.

(B) are all negative. This aligns with the negative coefficients of trade shock and female-intensive shock on the number of children per woman. Altogether, they might be indicative (though weakly evidenced) of fertility and sex-ratio trade-off: more sex selection among mothers with fewer children.

6.2 Robustness checks

To check for the robustness of our main results, we perform the following exercises:

Placebo tests: First, we run placebo (or permutation) tests. To do so, we randomly re-assign the level of tariff-cut exposure $(\Delta \tau_p)$ across provinces and run each of the estimations in the main analysis a thousand times. In any of these exercises, the estimated coefficients have normal distributions with the mean value centered around 0, suggesting no effects of the reshuffled treatments on the outcome variables. We present the graphs of the distribution of these estimates in the Appendix (Figures A.4 and A.5).

Estimations with complete census data: The data used in our main analysis

is only a fraction of the full population of the Vietnamese Census. To check if this sample selection drives our results, we re-run the main analysis using the full population census sample in 1999, 2009, and 2019. The full censuses contain less information than the short censuses: they only provide information about the household composition (age, gender, relation to the household head, years of education, ethnicity, rural/urban, and marital status). We check two main outcomes: (i) the share of girls among children under 5 years old at the provincial level and (ii) the number of children per household. Though there is no information about the fertility of the woman, we substitute by examining the impact of BTA on the average number of children per family (i.e., by counting the number of household members who are children of the household head and restricting the age of the mother to be under 35 or 40 years old). Our results are quite similar to the main results (Table B.4): BTA is not statistically associated with the share of female children under 5 years old and the average number of children per household across provinces.

The role of migration: Migration across provinces constitutes about 2% of the population in Vietnam.²⁰ We define migrants as individuals whose place of residence five years prior to the survey is different from their current place of residence. As trade shock creates better jobs in the provinces exposed to higher tariff cuts, it might induce more immigrants in these provinces. To assess if the effects are driven by migration, we re-run the analysis on a sample of non-migrants. The estimations on the sample of non-migrants are similar to the results in the main analysis (see Tables B.1 and B.2).

7 Mechanisms

In the previous Section, we find that the U.S.-Vietnam BTA has significantly induced a delay in marriage and fertility of young women, but it did not affect sex selection. In this Section, we explore possible mechanisms behind these effects by examining three channels: (1) the impacts of BTA on employment and sectoral transition of female workers; (2) the influence of BTA on the shift towards more female-intensive industries or increased segregation in the labor market; and (3) the effect of BTA on the educational attainment of women.

7.1 BTA and employment and structural change

We first examine if trade liberalization creates more jobs or alters the nature of women's jobs. In either case, it would affect the opportunity costs of getting married or having children. In this subsection, we estimate similar models as in equation 2 to analyze the following outcomes: women's employment status (whether employed or not) and the probabilities of women working in agriculture, manufacturing, and services.

 $^{^{20}}$ Own calculation from the population censuses

	working	agriculture	manufacturing	services
	(1)	(2)	(3)	(4)
$\Delta \tau_p \times 2009$	-0.003	-0.094***	0.071***	0.022*
-	(0.008)	(0.009)	(0.013)	(0.013)
$\Delta \tau_p \times 2019$	-0.001	-0.067***	0.053^{**}	0.014
-	(0.008)	(0.012)	(0.022)	(0.015)
Mean Y	0.795	0.483	0.192	0.325
Ν	6.070.753	4,951,395	4,951,395	4.951.395

Table 6: Impacts of BTA on employment and structural transformation, women aged 18-49

Notes: the estimations are on sample of women aged 18-49, also include the women's age, age-squared, ethnicity, and the province's based line characteristics in 1999 interacted with a linear time trend, province and census year fixed effects. Standard errors are clustered at province level and in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01.

Table 6 shows no statistically significant impact of BTA on women's employment This may be due to Vietnam's already high rate of women's labor force status. participation before trade liberalization. However, the coefficients of trade shocks are all negative, which could suggest that women stay in school longer and are, therefore, less likely to work. Columns (2) to (4) indicate a shift of women from agricultural to manufacturing jobs. This pattern of transformation is also observed among male workers (Table B.3). Our results are consistent with those of Asghar and McCaig (2024), which examines the impacts of BTA on sectoral transitions for individuals aged 15 to 29 years. Jobs in the manufacturing sector are more likely to offer higher salaries, be more formal, and have less flexible hours than those in agriculture. McCaig and Pavcnik (2018) find that BTA facilitates the transition from informal to formal sectors in Vietnam. These changes in the nature of jobs likely contribute to changes in gender norms and the empowerment of women, thereby influencing decisions related to marriage and fertility (Li, 2021; Rodríguez-Planas and Tanaka, 2022; Lee et al., 2021). Our results align with findings by Heath and Mobarak (2015) and Kis-Katos et al. (2018), who find that improved employment opportunities in the manufacturing sector lead to delayed marriage among women.

7.2 BTA and gender allocation in the labor market

In this subsection, we examine if BTA affects the relative allocation of males and females across sectors in the labor market. If female-intensive industries expand more rapidly than male-intensive industries, the economic position of women could improve significantly, potentially leading to a reduction or delay in marriage and fertility among women. In another aspect, even if trade creates equal employment opportunities for men and women, it might also trigger gender segregation in the labor market, with men and women becoming concentrated in different economic sub-sectors. Such segregation could reduce opportunities for matching between men and women, which could have significant consequences for marriage and fertility. Following Kis-Katos et al. (2018) and Ouyang et al. (2023), we construct two measures to assess these dynamics: the Female Intensity Index (FI) and the Gender Concentration Index (GCI).²¹ The Female Intensity Index is measured as follows:

$$FI_{pt} = \sum_{i} \left(\frac{L_{ipt}}{L_{pt}} \times f_{ip_0} \right) \tag{7}$$

In this equation, $\frac{L_{ipt}}{L_{pt}}$ represents the share of labor in sector *i* in province *p* relative to the total labor force of province *p* at time *t*; f_{ip_0} is the share of female worker in sector *i* in province *p* at baseline year (1999). An increase in *FI* indicates that the industry structure in province *p* is shifting towards more female-intensive industries at time *t*, given the initial gender structure of that province at the baseline.

The Gender Concentration Index is defined as:

$$GCI_{pt} = \sum_{i} \frac{L_{ip_0}}{L_{p_0}} \times \left(f_{ipt}^2 + (1 - f_{ipt})^2\right)$$
(8)

In equation 8, $\frac{L_{ip_0}}{L_{p_0}}$ represents the share of employment of industry *i* in province *p* relative to the total employment of province *p* at the baseline; f_{ipt} denotes the share of female workers in industry *i* in province *p* at time *t*; $1 - f_{ipt}$ is the share of male labors in the same industry and location. A higher value of *GCI* indicates greater gender segregation across sectors.²²

We then estimate equation 6 and replace the outcome variables by FI_{pt} and GCI_{pt} . Results of the estimations are presented in Table 7. We do not observe statistically significant impacts of BTA on either FI or GCI. This indicates that trade liberalization does not expand employment for women more relative to men or increase gender segregation across provinces. Therefore, we rule out these channels to explain the impacts of the U.S-Vietnam BTA on women's marriage and fertility in Vietnam.

7.3 BTA and the women's educational attainment

Another potential mechanism by which trade liberalization affects women's marriage and fertility is through its impact on women's educational attainment. More educated women not only spend longer at school but might also face higher opportunity costs of having children (Becker, 1960). Trade liberalization can increase income in

 $^{^{21}}$ We use the information on the sector of employment detailed at a 3-digit level and harmonize the industry classification in 2007 (Census 2009) and 2018 (Census 2019) to conform with the classification in 1993 (Census 1999).

²²The addition $f_{ipt}^2 + (1 - f_{ipt})^2$ gets the lowest value if $f_{ipt} = 0.5$

	female intensity index	gender concentration index
$\Delta \tau_p \times 2009$	-0.0841	-0.000479
	(-0.23)	(-0.24)
$\Delta \tau_p \times 2019$	0.0225	0.000136
-	(0.06)	(0.05)
Mean Y	46.982	0.562

Table 7: BTA and gender allocation in the labor market

Notes: N=180. The estimations also include province's based line characteristics in 1999 interacted with a linear time trend, province and census year fixed effects. Standard errors are clustered at province level and in parentheses (with bootstrap procedure). *p < 0.10, **p < 0.05, ***p < 0.01.

regions more exposed to trade, making it more affordable for families in these regions to invest in their children's education. McCaig (2011) finds a higher reduction in the poverty rate in provinces exposed more to tariff reduction within four years after the implementation of BTA. In addition, McCaig and Pavcnik (2018) find that the U.S.-Vietnam BTA promotes the reallocation of labor from the informal to the formal markets, where higher skill levels are often required. Given that women would experience higher returns to investment in education compared to men (Pitt et al., 2012), they are more incentivized to pursue further schooling.

We estimate equation 2, with the dependent variables being the total years of education of the women at the time of the survey. The results are presented in Table 8. Column (1) shows that a one standard deviation (1.37 percentage points) increase in the province's exposure to tariff reduction is associated with an increase in women's average years of schooling by 0.24 years by 2009 and by 0.26 years by 2019. The positive impacts are particularly evident among the younger cohorts of women (aged 18-28 and 29-39 years old) and among the older cohort in 2019. Heath and Mobarak (2015) also find significant effects of exposure to garment jobs to the educational attainment and school enrollment of both boys and girls.

8 Conclusions

In this paper, we study the impacts of a positive trade shock on the marriage, fertility, and sex ratio in Vietnam over the period of nearly 20 years after the shock. We find that the shock significantly reduces marriage and fertility rate among young women (aged 18-28), implying a delay or postponement of marriage and reproductive behaviors. We observe virtually no effect of the shock on sex-selective behavior in Vietnam. Our analysis further shows that the trade shock created by the implementation of the U.S.-Vietnam BTA is relatively gender-neutral, which promotes the transition from agriculture to manufacturing for both men and women without inducing greater gender

	18-49	18-28	29-39	40-49
	(1)	(2)	(3)	(4)
$\Delta \tau_p \times 2009$	0.178^{***}	0.191^{***}	0.226^{***}	0.080
	(0.058)	(0.054)	(0.080)	(0.056)
$\Delta \tau_p \times 2019$	0.191^{***}	0.111*	0.245^{***}	0.226^{***}
1	(0.062)	(0.066)	(0.079)	(0.067)
Mean Y	8.83	9.56	8.81	7.78
Ν	6.070.197	2.261.477	2.157.007	1.651.713

Table 8: Impacts of BTA on years of education

Notes: the estimations also include the women's age, age-squared, ethnicity, and the provinces's based line characteristics in 1999 interacted with a linear time trend, province and census year fixed effects. Standard errors are clustered at province level and in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01.

segregation or disproportionately expanding female-intensive industries across provinces. This distinguishes Vietnam from other countries examined in the existing literature on trade liberalization, which often find disproportional impacts on females and males in the labor market. Due to the high share of women's participation in the labor market at baseline, this shock does not affect women's employment, at least at the extensive margin. The impacts of this shock on marriage and fertility are driven by changing the nature of women's jobs and improving women's education.

We acknowledge that our analyses have limitations. While our study does not observe the relative expansion of female-intensive industries or effects of the trade shock on gender segregation in the labor market, we are unable to test whether this shock alters the relative earnings of women compared to men - a key channel that could change the opportunity costs of marriage and having children. Unfortunately, the population census data available to us does not provide information on individual income. In addition, our study does not offer a definite explanation for the increasing sex ratio at birth in Vietnam. Many studies have shown that families with better socio-economic status, such as those with more educated mothers and higher wealth quintiles, are more likely to engage in sex-selection practices (Bhalotra and Cochrane, 2010; UNFPA, 2019). In this paper, we show that trade liberalization resulting from the U.S.-Vietnam BTA promotes structural transformation (which often leads to increased income) and women's educational attainment. However, we do not find significant effects of BTA on sex-selective behavior. A possible explanation is that son preference is deeply embedded in the country's culture, and regions with stronger son preference might not coincide with those experiencing greater exposure to the BTA's economic impacts.

References

- Anukriti, Sharma and Todd J Kumler, "Women's worth: Trade, female income, and fertility in India," *Economic Development and Cultural Change*, 2019, 67 (3), 687–724.
- Asghar, Ramisha and Brian McCaig, "Trade, structural change, and labour market transitions in Vietnam," Technical Report, STEG Working paper no. 093 2024.
- Autor, David, David Dorn, and Gordon Hanson, "When work disappears: Manufacturing decline and the falling marriage market value of young men," *American Economic Review: Insights*, 2019, 1 (2), 161–178.
- Autor, David H, David Dorn, and Gordon H Hanson, "The China syndrome: Local labor market effects of import competition in the United States," American Economic Review, 2013, 103 (6), 2121–2168.
- _ , _ , _ , and Jae Song, "Trade adjustment: Worker-level evidence," The Quarterly Journal of Economics, 2014, 129 (4), 1799–1860.
- Babiarz, Kimberly Singer, Paul Ma, Grant Miller, Shige Song et al., The limits (and human costs) of population policy: Fertility decline and sex selection in China under Mao, National Bureau of Economic Research, 2018.
- Banerji, Angana, Albe Gjonbalaj, Sandile Hlatshwayo, and Anh Van Le, "Asian women at work," Finance & Development. Available from URL: https://search. proquest. com/docview/2112561819, 2018.
- Becker, Gary S, "An economic analysis of fertility," in "Demographic and economic change in developed countries," Columbia University Press, 1960, pp. 209–240.
- _, A Treatise on the Family, Havard University Press, 1981.
- _ , "Human capital, effort, and the sexual division of labor," *Journal of Labor Economics*, 1985, 3 (1, Part 2), S33–S58.
- and H Gregg Lewis, "Manufacturing growth between the quantity and quality of children," Journal of Political Economy, 1973, 81 (2, Part 2), S279–S288.
- Besedeš, Tibor, Seung Hoon Lee, and Tongyang Yang, "Trade liberalization and gender gaps in local labor market outcomes: Dimensions of adjustment in the United States," *Journal of Economic Behavior & Organization*, 2021, 183, 574–588.
- Bhalotra, Sonia R and Tom Cochrane, "Where have all the young girls gone? Identification of sex selection in India," Technical Report, IZA Discussion Papers 2010.
- Boldrin, Michele and Larry E Jones, "Mortality, fertility, and saving in a Malthusian economy," *Review of Economic Dynamics*, 2002, 5 (4), 775–814.
- Borusyak, Kirill, Peter Hull, and Xavier Jaravel, "Quasi-experimental shift-share research designs," *The Review of Economic Studies*, 2022, 89 (1), 181–213.
- Braga, Breno, "The effects of unilateral trade liberalization on marriage and fertility choices: Evidence from Brazil," Available at SSRN 3161625, 2018.
- Brussevich, Masha, "Does trade liberalization narrow the gender wage gap? The role

of sectoral mobility," European Economic Review, 2018, 109, 305–333.

- Chakraborty, Tanika, "Trade liberalization in a traditional society: implications for relative female survival," World Development, 2015, 74, 158–170.
- Chatterjee, Shoumitro and Tom Vogl, "Escaping Malthus: economic growth and fertility change in the developing world," *American Economic Review*, 2018, 108 (6), 1440–1467.
- **Dix-Carneiro, Rafael and Brian K Kovak**, "Trade liberalization and the skill premium: A local labor markets approach," *American Economic Review*, 2015, 105 (5), 551–557.
- and _ , "Trade liberalization and regional dynamics," American Economic Review, 2017, 107 (10), 2908–46.
- Do, Quy-Toan, Andrei A Levchenko, and Claudio Raddatz, "Comparative advantage, international trade, and fertility," *Journal of Development Economics*, 2016, 119, 48–66.
- **Doepke, Matthias**, "Accounting for fertility decline during the transition to growth," Journal of Economic Growth, 2004, 9 (3), 347–383.
- _, "Gary Becker on the quantity and quality of children," Journal of Demographic Economics, 2015, 81 (1), 59–66.
- Ebenstein, Avraham and Steven Leung, "Son preference and access to social insurance: Evidence from China's rural pension program," *Population and Development Review*, 2010, 36 (1), 47–70.
- Engel, Jakob, Deeksha Kokas, Gladys Lopez-Acevedo, and Maryla Maliszewska, The distributional impacts of trade: Empirical innovations, analytical tools, and policy responses, World Bank Publications, 2021.
- Erten, Bilge and Jessica Leight, "Exporting out of agriculture: The impact of WTO accession on structural transformation in China," *Review of Economics and Statistics*, 2021, 103 (2), 364–380.
- _ , _ , and Lianming Zhu, "The Effects of FDI Liberalization on Structural Transformation and Demographic Change: Evidence from China," Available at SSRN 4614188, 2023.
- Feng, Qiyangfan, Yiwen Shangguan, and Hao Wei, "Does opening up change customs? Evidence from trade liberalization and marriage payments in China," *Review of Development Economics*, 2024.
- Fukase, Emiko, "Export liberalization, job creation, and the skill premium: Evidence from the US-Vietnam bilateral trade agreement (BTA)," World Development, 2013, 41, 317–337.
- Gaddis, Isis and Janneke Pieters, "The gendered labor market impacts of trade liberalization: Evidence from Brazil," *Journal of Human Resources*, 2017, 52 (2), 457–490.
- Giuntella, Osea, Lorenzo Rotunno, and Luca Stella, "Globalization, fertility, and marital behavior in a lowest-low fertility setting," *Demography*, 2022, 59 (6), 2135–2159.

- Goldsmith-Pinkham, Paul, Isaac Sorkin, and Henry Swift, "Bartik instruments: What, when, why, and how," *American Economic Review*, 2020, 110 (8), 2586–2624.
- Gupta, Ashmita, "Effect of trade liberalization on gender inequality: the case of India," *IMF Economic Review*, 2021, 69 (4), 682–720.
- Gupta, Monica Das and PN Mari Bhat, "Fertility decline and increased manifestation of sex bias in India," *Population Studies*, 1997, 51 (3), 307–315.
- Heath, Rachel and A Mushfiq Mobarak, "Manufacturing growth and the lives of Bangladeshi women," Journal of Development Economics, 2015, 115, 1–15.
- Huynh, Nghiem and Ngoc Nguyen, "Why does the sex ratio at birth rise? Evidence from Vietnam," Technical Report, http://dx.doi.org/10.2139/ssrn.4582123 2023.
- Iqbal, Neelam, Anna Gkiouleka, Adrienne Milner, Doreen Montag, and Valentina Gallo, "Girls' hidden penalty: Analysis of gender inequality in child mortality with data from 195 countries," *BMJ Global Health*, 2018, 3 (5), e001028.
- Jayachandran, Seema, "Fertility decline and missing women," American Economic Journal: Applied Economics, 2017, 9 (1), 118–139.
- Juhn, Chinhui, Gergely Ujhelyi, and Carolina Villegas-Sanchez, "Trade liberalization and gender inequality," American Economic Review, 2013, 103 (3), 269–273.
- Keller, Wolfgang and Hale Utar, "Globalization, gender, and the family," The Review of Economic Studies, 2022, 89 (6), 3381–3409.
- Kis-Katos, Krisztina, Janneke Pieters, and Robert Sparrow, "Globalization and social change: Gender-specific effects of trade liberalization in Indonesia," *IMF Economic Review*, 2018, 66, 763–793.
- Kovak, Brian K, "Regional effects of trade reform: What is the correct measure of liberalization?," American Economic Review, 2013, 103 (5), 1960–76.
- Lee, Bun Song, Jennifer Klein, Mark Wohar, and Sangsin Kim, "Factors delaying marriage in Korea: An analysis of the Korean population census data for 1990–2010," Asian Population Studies, 2021, 17 (1), 71–93.
- Li, Jie, "Women hold up half the sky? Trade specialization patterns and work-related gender norms," *Journal of International Economics*, 2021, 128, 103407.
- Luo, Wei and Xianqiang Zou, "Demographic impacts of China's trade liberalization: marriage, spousal quality, and fertility," *Journal of Population Economics*, 2024, 37 (3), 1–31.
- Mansour, Hani, Pamela Medina, and Andrea Velásquez, "When Women's Work Disappears: Marriage and Fertility Decisions in Peru," *Journal of Globalization and Development*, 2023, 14 (2), 385–412.
- McCaig, Brian, "Exporting out of poverty: Provincial poverty in Vietnam and US market access," *Journal of International Economics*, 2011, 85 (1), 102–113.
- and Nina Pavcnik, "Export markets and labor allocation in a low-income country," American Economic Review, 2018, 108 (7), 1899–1941.
- _ , _ , and Woan Foong Wong, "FDI inflows and domestic firms: Adjustments to

new export opportunities," Technical Report, National Bureau of Economic Research 2022.

- Morrow, Peter M, "Ricardian–Heckscher–Ohlin comparative advantage: Theory and evidence," Journal of International Economics, 2010, 82 (2), 137–151.
- Ngo, Anh P, "Effects of Vietnam's two-child policy on fertility, son preference, and female labor supply," *Journal of Population Economics*, 2020, pp. 1–44.
- Ngoc, Pham Phuong, Hanol Lee, and Dainn Wie, "Trade liberalization and women's marriage and fertility decisions in Vietnam," *Economic Analysis and Policy*, 2024, 82, 1071–1082.
- Nugent, Jeffrey B, "The old-age security motive for fertility," *Population and Development Review*, 1985, pp. 75–97.
- Ouyang, Difei, Weidi Yuan, and Yuan Zi, "Empowered young women: trade liberalization and women's family decisions in china," Technical Report, CEPR Discussion paper no. 17607 2023.
- Pitt, Mark M, Mark R Rosenzweig, and Mohammad Nazmul Hassan, "Human capital investment and the gender division of labor in a brawn-based economy," *American Economic Review*, 2012, 102 (7), 3531–3560.
- Qian, Nancy, "Missing women and the price of tea in China: The effect of sex-specific earnings on sex imbalance," *The Quarterly Journal of Economics*, 2008, 123 (3), 1251–1285.
- **Ricardo, David**, "From the principles of political economy and taxation," in "Readings in the economics of the division of labor: The classical tradition," World Scientific, 2005, pp. 127–130.
- **Riedel, James, Steve Parker, and Phan Vinh Quang**, An Assessment of the Economic Impact of the United States-Vietnam Bilateral Trade Agreement, The National Politial Publishing House, Hanoi, 2003.
- Rodríguez-Planas, Núria and Ryuichi Tanaka, "Gender norms and women's decision to work: Evidence from Japan," *Review of Economics of the Household*, 2022, 20 (1), 15–36.
- Rossi, Pauline and Mathilde Godard, "The old-age security motive for fertility: Evidence from the extension of social pensions in Namibia," *American Economic Journal: Economic Policy*, 2022, 14 (4), 488–518.
- Siddiqui, Kalim, "Trade liberalization and economic development: A critical review," International Journal of Political Economy, 2015, 44 (3), 228–247.
- ____, "David Ricardo's comparative advantage and developing countries: Myth and reality," International Critical Thought, 2018, 8 (3), 426–452.
- Stevenson, Betsey and Justin Wolfers, "Marriage and divorce: Changes and their driving forces," Journal of Economic Perspectives, 2007, 21 (2), 27–52.
- Tafuro, Sara and Christophe Z Guilmoto, "Skewed sex ratios at birth: A review of global trends," *Early Human Development*, 2020, 141, 104868.
- Topalova, Petia, "Trade liberalization, poverty and inequality: Evidence from Indian

districts," in "Globalization and Poverty," University of Chicago Press, 2007, pp. 291–336.

- _, "Factor immobility and regional impacts of trade liberalization: Evidence on poverty from India," *American Economic Journal: Applied Economics*, 2010, 2 (4), 1–41.
- **UNFPA**, "Sex ratio at birth in Vietnam: New evidence from the Intercensal Population and Housing Survey in 2014," Technical Report, United Nations Population Fund in Vietnam 2014.
- _ , "Sex imbalances at birth in Vietnam: Trends and variations Analysis from the 2019 Population and Housing Census," Technical Report, United Nations Population Fund in Vietnam 2019.
- Vu, Linh Hoang, Tuyen Quang Tran, and Tung Duc Phung, "Children and female labor market outcomes in Vietnam," *Heliyon*, 2021, 7 (7).
- Vu, Tien Manh, "One male offspring preference: Evidence from Vietnam using a split-population model," *Review of Economics of the Household*, 2014, 12 (4), 689–715.

A Figures



Figure A.1: Share of US in total VN's exports *Source*: UNCOMTRADE.



Figure A.2: Exports from Vietnam to the U.S and other markets, 1995-2010 $_{Source:}$ COMTRADE



Figure A.3: Provincial exposure to tariff cuts by the U.S. - Vietnam BTA



Figure A.4: Distribution of the estimates of equation 2 on sample of women aged 18-28, dependent variable is the probability of being single, 1,000 times randomly assign the level of exposure to tariff reduction across provinces



Figure A.5: Distribution of the estimates of equation 2 on sample of women aged 18-28, dependent variable is the number of children, 1,000 times randomly assign the level of exposure to tariff reduction across provinces

B Tables

		1					1.11	
		being	single			no. of o	children	
	18-49	18-28	29-39	40-49	18-49	18-28	29-39	40-49
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Panel A								
$\Delta \tau \times 2009$	0.012**	0.022***	0.006	0.006	-0.040	-0.040**	-0.029	-0.041
	(0.005)	(0.006)	(0.006)	(0.004)	(0.039)	(0.016)	(0.052)	(0.070)
$\Delta \tau \times 2019$	0.019^{***}	0.035***	0.015^{**}	0.007	-0.035	-0.056***	-0.031	-0.003
	(0.005)	(0.005)	(0.006)	(0.004)	(0.039)	(0.012)	(0.052)	(0.077)
Panel B								
$\Delta \tau^f \times 2009$	0.029***	0.055***	0.012	0.016**	-0.074	-0.122***	-0.077	0.055
	(0.009)	(0.014)	(0.009)	(0.006)	(0.065)	(0.038)	(0.091)	(0.133)
$\Delta \tau^f \times 2019$	0.021**	0.045***	0.015	0.010	-0.018	-0.095**	-0.028	0.137
	(0.009)	(0.013)	(0.011)	(0.008)	(0.084)	(0.037)	(0.119)	(0.160)
$\Delta\tau^m\times 2009$	-0.015	-0.028	-0.002	-0.009	0.015	0.090^{*}	0.049	-0.183
	(0.010)	(0.018)	(0.010)	(0.007)	(0.068)	(0.050)	(0.102)	(0.164)
$\Delta \tau^m \times 2019$	0.017	0.021	0.014	0.003	-0.064	0.001	-0.036	-0.217
	(0.012)	(0.021)	(0.013)	(0.010)	(0.113)	(0.051)	(0.162)	(0.231)
N	5,766,305	2,055,842	2,084,457	1,626,006	5,766,305	2,055,842	2,084,457	1,626,006

Table B.1: Impacts of BTA on m	marriage and fertility,	non-migrants
--------------------------------	-------------------------	--------------

Notes: Sample of non migrants. The estimations also include the women's age, age-squared, ethnicity, provinces's based line characteristics in 1999 interacted with a linear time trend, province and census year fixed effects. Standard errors are clustered at province level and in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01.

		provincial level			
	18-49	18-28	29-39	40-49	
	(1)	(2)	(3)	(4)	(5)
Panel A					
$\Delta \tau_p \times 2009$	-0.0001	-0.0019	0.0017	-0.0013	-0.0008
1	(0.0017)	(0.0022)	(0.0012)	(0.0033)	(0.0010)
$\Delta \tau_p \times 2019$	0.0007	-0.0011	0.0016	0.0002	-0.0012
1	(0.0019)	(0.0023)	(0.0012)	(0.0035)	(0.0012)
Panel B					
$\Delta \tau_p^f \times 2009$	0.0031	0.0005	0.0003	0.0090	-0.0018
1	(0.0039)	(0.0052)	(0.0033)	(0.0064)	(0.0023)
$\Delta \tau_p^f \times 2009$	0.0036	0.0059	-0.0021	0.0107	-0.0001
1	(0.0048)	(0.0061)	(0.0035)	(0.0080)	(0.0028)
$\Delta \tau_p^m \times 2009$	-0.0051	-0.0056	0.0037	-0.0170*	0.0006
-	(0.0058)	(0.0071)	(0.0060)	(0.0089)	(0.0033)
$\Delta \tau_p^m \times 2019$	-0.0038	-0.0123	0.0074	-0.0158	-0.0031
-	(0.0075)	(0.0088)	(0.0066)	(0.0120)	(0.0038)
Mean Y	0.443	0.464	0.444	0.429	0.478
N	4,316,978	968,907	$1,\!855,\!024$	$1,\!493,\!047$	180

Table B.2: Impacts of BTA on sex selection, non-migrants

Notes: Sample of non-migrants. In Column (5), we only consider children under 5 years old in households whose household heads are non-migrants. The estimations also include the women's age, age-squared, ethnicity, province's based line characteristics in 1999 interacted with a linear time trend, province and census year fixed effects. Standard errors are clustered at province level and in parentheses, and with bootstrap procedure in Column (5). *p < 0.10, **p < 0.05, ***p < 0.01.

Table B.3: Impacts of BTA on employment and structural transformation, male aged 18-49

	working	agriculture	manufacturing	services
	(1)	(2)	(3)	(4)
$\Delta \tau_p \times 2009$	-0.006	-0.085***	0.051^{***}	0.034^{***}
	(0.005)	(0.007)	(0.009)	(0.010)
$\Delta \tau_p \times 2019$	-0.007	-0.063***	0.041**	0.022^{*}
-	(0.006)	(0.010)	(0.016)	(0.012)
Mean Y	0.882	0.463	0.154	0.382
Ν	5,979,647	5,336,299	5,336,299	5,336,299

Notes: the estimations are on sample of men aged 18-49, also include the women's age, age-squared, ethnicity, and the provinces's based line characteristics in 1999 interacted with a linear time trend, province and census year fixed effects. Standard errors are clustered at province level and in parentheses. *p < 0.10, **p < 0.05, ***p < 0.01.

	share of girls		no. of children		no. of children	
	children under 5		mother under 35		mother under 40	
	(1)	(2)	(3)	(4)	(5)	(6)
$\Delta \tau_p \times 2009$	-0.000332		-0.0352*		-0.0341	
	(-0.62)		(-1.66)		(-1.50)	
$\Delta \tau_p \times 2019$	0.000173		-0.0343		-0.0267	
-	(0.28)		(-1.27)		(-0.89)	
$\Delta \tau_p^f \times 2009$		-0.00268*		-0.0429		-0.0133
1		(-1.95)		(-0.82)		(-0.23)
$\Delta \tau_p^f \times 2019$		-0.00243		-0.00827		0.0351
Ĩ		(-1.09)		(-0.12)		(0.47)
$\Delta \tau_p^m \times 2009$		0.00323^{*}		-0.0259		-0.0693
1		(1.70)		(-0.37)		(-0.91)
$\Delta \tau_p^m \times 2019$		0.00403		-0.0744		-0.122
Ľ		(1.22)		(-0.77)		(-1.25)
N	180	180	180	180	180	180

Table B.4: Impacts of BTA on child sex ratio and fertility, full census 1999-2019

Notes: The estimations are applied on the full population census in 1999, 2009, 2019. The estimations include census year and province fixed effects. Standard errors are clustered at the provincial level with bootstrap procedure. We do not include any other independent variables. *p < 0.10, **p < 0.05, ***p < 0.01.